**Author’s response to reviews**

**Title:** Differential expression of connective tissue growth factor and hepatocyte growth factor in the vitreous of patients with high myopia versus vitreomacular interface disease

**Authors:**

Xinyi Ding (dr_xding@outlook.com)

Rong Zhang (happyreny@sina.com)

Shujie Zhang (13881465@163.com)

Hong Zhuang (zhuang_hong1008@126.com)

Gezhi Xu (drxugezhi@163.com)

**Version:** 2  **Date:** 12 Jan 2019

**Author’s response to reviews:**

Dear editor and reviewers,

Thank you very much for your advice. We have revised the manuscript, and would like to re-submit it for your consideration. According to the editor’s requests, we have made a few minor revisions to the manuscript. The minor amendments included: the Ethics approval and consent to participate section/ the Availability of data and materials section/ the Authors’ contributions section.

We sincerely hope that editor and reviewers will be satisfied with this revised version of the manuscript.